Author's response to reviews

Title: Aeromonas sobria necrotizing fasciitis and sepsis in an immunocompromised patient: case report and literature review.

Authors:

Savino Spadaro (savinospadaro@gmail.com)
Angela Berselli (angela.berselli@libero.it)
Elisabetta Marangoni (e.marangoni@ospfe.it)
Anna Romanello (a.romanello@ospfe.it)
Maria V Colamussi (mv.colamussi@ospfe.it)
Riccardo Ragazzi (rgc@unife.it)
Silvia Zardi (s.zardi@ospfe.it)
Carlo A Volta (vlc@unife.it)

Version: 3
Date: 8 April 2014

Author's response to reviews: see over
Dear editor in chief Michael Kidd,

On behalf of my co-authors, I am submitting the enclosed material for possible publication in the *Journal of Medical case report*. It has not been submitted for publication nor has it been published in whole or in part elsewhere. I attest to the fact that all authors listed on the title page have read the manuscript, attest to the validity and legitimacy of the data and its interpretation, and agree to its submission *Journal of Medical case report*. I acknowledge that both I and the other authors have read the Instructions for Authors and agree with its contents.

We hope that you will find this case interesting. We tried to describe an uncommon case of necrotizing fasciitis caused by *Aeromonas sobria* and progressive sepsis in a immunocompromised patient and to emphasize that difficult recognition, management and aggressive treatment are the primary determinants of outcome for patients with skin and soft tissue necrotizing infection.

Regarding the COMPETING INTERESTS, the authors declare that they have no competing interests.

Yours sincerely.

Ferrara, 01/04/2014

Prof. CARLO ALBERTO VOLTA